Author’s response to reviews

Title: Mutations in the PIGW gene associated with hyperphosphatasia and mental retardation syndrome: a case report

Authors:

Lina Fu (fulina1989927@163.com)
Yan Liu (lyan3022@163.com)
Yu Chen (1030756510@qq.com)
Yi Yuan (yuanyimed@163.com)
Wei Wei (wei.wei@kangso.net)

Version: 8 Date: 13 Feb 2019

Author’s response to reviews:

Editor Comments:

1. Thank you for the clarification provided in your email correspondence. We ask that you take this opportunity to revise the language in your ‘Ethics approval and consent to participate’ subsection so as to clarify that the parents of the patient provided their written informed consent to participate in genetic testing.

Reply: We have clarified that the parents of the patient provided their written informed consent to participate in genetic testing in the ‘Ethics approval and consent to participate’ subsection in line 23, page 6.